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Nr Item Responsible Comment Status 

1 Procedure for galleries access 
B. Rae, J. 

Bernhard  

New version under approval and to be merged with 

EDMS 1748425. EDMS 2391345 
Ongoing 

2 
BE-DSO recommendation for 

audible alarm in GIF++ 

M. Jaekel, N. 

Charitonidis 
To be clarified with BE-DSO. Ongoing 

https://edms.cern.ch/document/SPSX-S-OP-0001/0.3/approvalAndComments


   

 

   
 

NA-CONS Related Activities 
 

Nr Item Responsible Comment Status 

1 
ATEX ventilation 

works 
EN-CV 

 VIC foreseen on Monday, 12.02.24, works will 

follow. It will be finished by the end of March 
Ongoing 

2 

Consolidation of the gas 

mixing area on the Jura 

side 

BE-EA Activity postponed to YETS 2024/25. Postponed 

3 

Consolidation of the 

chilled water 

piping/distribution in 

EHN1 

EN-CV 
Advancing slowly, 30% done. End date remains 22nd 

March (wk12) 

2023/24 

YETS 

4 

EHN2 Underground 

Galleries Ventilation 

Installation    

EN-CV  
In progress: Major works completed. Commissioning 

by the end of February. 
Ongoing 

5 
Installation of the fire 

doors in GHN21/22 
EN-ACE 

In progress: Works on-going. 50 % of work 
completed for both BA81-EHN1 & 

EHN2/GHN2. Completion foreseen in wk9. 
Ongoing 

6 
Installation/test of new 

rad-hard monitor 
BE-EA / SY-BI  

TT83: All necessary works have been done. Beam 

pipe resizing, cabling, fibre optics, and installing 

support. Installation begins next week (12th Feb). 

Ongoing 

7 

Consolidation of the fire 

sprinkler station in 

BA80/PA80/TA802 

EN-AA 

Shaft PA80: the two vertical lines are installed. 

Starting with the antennas on each floor. 

Connection of the new sprinkler station to the 

old system for the PGT802 shaft is ongoing. 

Issues to be clarified. Postponed to 

Ongoing 



   

 

   
 

 

EYETS24/25: sprinkler installation in TA801, 

PGT802 and GL802. 

8 
Fire detection 

underground 
EN-AA 

 TT81, TT82, TT83, and TT85 are completed 

(speaker and Bris de glace), Left to install 7 

"boxes" in TT84 and move the speakers.  Cable 

trays are nearly completed for all TTs. Cabling 

is ongoing and completion is foreseen in wk6. 

Ongoing 

9 
Alignment of H6/H8 

beamlines 
BE-GM 

Start was delayed by 3 weeks, works are 

ongoing and 90% completed. Foreseen to be 

completed by week 7. 

Ongoing 


